Author’s response to reviews

Title: THE ITALIAN VERSION OF THE CRANIOFACIAL PAIN AND DISABILITY INVENTORY IN PATIENTS WITH CHRONIC TEMPOROMANDIBULAR JOINT DISORDERS. Cross-cultural adaptation, reliability and validity

Authors:

Marco Monticone (mmonticone@gmail.com)
Barbara Rocca (barbara.rocca@icsmaugeri.it)
Paola Abelli (paola.abelli@icsmaugeri.it)
Simona Tecco (tecco.simon@hsr.it)
Tommaso Geri (marcotesta.unige@gmail.com)
Enrico Gherlone (gheron.enrico@hsr.it)
Deborah Luzzi (deborah.luzzi@gmail.com)
Marco Testa (marco.testa@unige.it)

Version: 3 Date: 08 Oct 2019

Author’s response to reviews:

Ref.: Ms. OHEA-D-18-00171R2 – BMC Oral Health

THE ITALIAN VERSION OF THE CRANIOFACIAL PAIN AND DISABILITY INVENTORY IN PATIENTS WITH CHRONIC TEMPOROMANDIBULAR JOINT DISORDERS. Cross-cultural adaptation, reliability and validity

Dear Editor,

We addressed all of your requests and our point-by-point replies are given below.

Yours faithfully,

The Authors
Editorial requests

1. Authorship change form

-- We note that the order of authors has changed since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

In such cases, we use a standardised form which we would request you and your co-authors to complete. The authorship change form can be found at the following link: https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5.

All instructions can be found on the form, please treat the 'current authorship' section as the original authorship. Please return the form within 14 days by email to the Editorial Office with all author signatures (including those newly added/removed).

Answer: The form will be sent as soon as possible to the editorial office as requested.

2. Consent to participate

-- In your “ethical approval and consent to participate” section, please confirm whether informed consent was obtained in written or verbal format. If verbal, please confirm if an ethics committee approved the procedure for verbal consent, and why they felt the need for written consent was not necessary for this study. Please also detail how you documented the consent.

Answer: The consent was obtained as written consent to participate to the study. This specification was added into the relative section in the declaration at the end of the manuscript.

3. Availability of data and materials

-- The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based.

Answer: we reworded the phrase as follow: “All relevant data generated or analysed during this study are included in this published article. Raw data are available on request to authors”
4. Consent for publication

-- Please amend your “Consent for publication” section in your Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

Answer: done

5. Authors' contributions

-- Please clarify whether the initials 'TM' and 'RB' indicate the authors Marco Testa and Barbara Rocca.

Answer: Yes, there was a mistake in the order of first/last name for the initials. We have corrected. Now Marco Testa appears as MT and Barbara Rocca as BR in the text

6. Additional files

-- Please ensure that both additional files are cited within the main text. Any items which are not cited may be deleted by our production department upon publication.

Answer: We have put the citation of the additional files in the manuscript where needed: line 13 of the Background section: “…For this purpose, in 2014 the Craniofacial Pain and Disability Inventory (CFPDI, see additional file 2 for Spanish version)…”

7. Clean manuscript

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Answer: Done